Author’s response to reviews

Title: Monocular lens dislocation due to vomiting-a case report

Authors:
Min Wang (angin2001@126.com)
Yu Fang Gao (gaoyf06@126.com)
Rong Li (15991684943@163.com)
Sheng Wang (635111324@qq.com)

Version: 2 Date: 09 Oct 2017

Author’s response to reviews:

Reviewer #1: The following issues should be considered;

1- Please clarify the exact setting for this case report.
The exact setting for this case report has been clarified.

2- Please added others investigations if done.
Other investigations were not done.

3- Discussion is not sufficient.
The discussion has been supplemented.

4- Conclusion should be rephrase.
The conclusion has been rephrased.

5- Some typographical errors should be corrected.
Some typographical errors has been corrected.
ELENA Pacella (Reviewer 2): I have to compliment the Authors because they actually described what was often the case in cataract operated patients in general anesthesia. The complication of vomiting induced by awakening from general anesthesia caused the IOL in anterior chamber (AC) to become more frequent.

Currently the advent of topical and/or local anesthetic techniques has reduced these complications.

- page 2 line 33 correct…….. Backgroud with Background

Some typographical errors has been corrected.

Background:


The reference has been added.

- page 2, line 55…………. Authors should better specify the text when this complication may occur more frequently in the postoperative period, or whether the complication may depend on the IOL construction material, or even considering that the patient was unaffected by any syndrome, dislocation of IOL could be secondary to complications of cataracts.

There was no history of trauma and surgery, so we concluded that lens dislocation in his right eye was caused by vomiting due to dizziness.

Case presentation:

- Page 3, line  22 ..... I think it is appropriate that the authors should better describe how long they have gone through cataract surgery and the acute vomiting episode that has led to the discovery of the lens into the anterior chamber.

The patient did not accept cataract surgery. He was diagnosed with "simply vomiting induced lens dislocation" and accepted surgical treatment of this rare condition after 48 hours.